Author's response to reviews

Title: Identifying Effective Retention Strategies: A Research Agenda

Authors:

Anna Kearney (a.kearney@liv.ac.uk)
Anne Daykin (anne.daykin@bristol.ac.uk)
Alison Shaw (ali.heawood@bristol.ac.uk)
Athene Lane (Athene.Lane@bristol.ac.uk)
Jane Blazeby (J.M.Blazeby@bristol.ac.uk)
Mike Clarke (m.clarke@qub.ac.uk)
Paula Williamson (prw@liv.ac.uk)
Carrol Gamble (c.gamble@liv.ac.uk)

Version: 1 Date: 29 Jun 2017

Author’s response to reviews:

A detailed response to the reviewer comments. Has been uploaded as a separate word document in the attached files.